% SIMIODE-TeacherStudent-Template.tex and simiode.cls for LaTeX2e
% Version 22 November 2016.
%
% This template is to be used with the  class file "simiode.cls".
%
% The latest version of this class and template files
% can be found at
%
%       <http://www.SIMIODE.org> in the Author section.
%
% A "Quick Start Guide" appears below.
%
% This file must be processed with version LaTeX2e or higher.
% Appreciation to Mark Frantz, IUPUI; John Thoo, Yuba College; and Eric Sullivan, Carroll College.
%
\documentclass{simiode}
%
% With this version of the tex file and the associated SIMIODE.cls file you can produce BOTH a Student Version and a Teacher Version of your modeling scenario or technique narrative.
%
% Student Version contains the title STUDENT VERSION along with the NAME OF SCENARIO, with no abstract, no keywords, no tags, no author identifiers, and no COMMENTS section. The Student Version contains only the STATEMENT of the modeling scenario only with appropriate figures, data, instructions, and perhaps a bibliography appropriate for student resourcing, but nothing in the bibliography which would "give away" the nature of the modeling scenario, etc.
%
%The Teacher Version contains the material in the Student Version, namely the title NAME OF SCENARIO, now headed with   TEACHER VERSION, but also full author  identification, abstract, key words, tag words, and, following the STATEMENT, a section of COMMENTS which can be discussion of presentation  ideas, "solutions," techniques, etc.
%
% Authors can prepare a full Teacher Version using the notions immediately below  to create a pdf for review, but in review please eliminate all information about the author as we use a double-blind referee system. Such information will be restored when accepted and published in SIMIODE at www.simiode.org.
%
%***********   Toggle the following line to turn the teacher version on (1) or off (0). ***********
%
\def\TeacherEdition{0} % Toggle this line's value.
\newcommand{\teacher}[1]{\ifnum\TeacherEdition=1 #1 \fi}
%
% The \teacher{ } command wraps any text that will appear only in the teacher version of
% the document. See this file for examples. This has been done for you in the document below.
% Rename the pdf after you compile each final version.
%
\usepackage{amsmath,amssymb}
\usepackage{color}
\usepackage{graphicx}
\usepackage{hyperref}
\usepackage{url}
%
\renewcommand{\baselinestretch}{1.3}
%
% NOTE regarding macros %%
%
% Any macros defined for your paper should be contained in
% the top matter. Likewise, any environment definitions such
% as \newtheorem or \newenvironment should also go in the
% top matter.
%
% Do not \input your macros as separate files within your
% final version because more files make it harder for the
% editors and users to keep track of and/or modify and
% append your paper.
%
% Top matter is above and the beginning of the document below.
%
\begin{document}
\begin{thebibliography}{98}


\bibitem{Ear} Mile high hearing. \href{https://milehighhearing.com/blog/types-of-hearing-loss}{https://milehighhearing.com/blog/types-of-hearing-loss}.  Accessed on 20 August 2018.



\bibitem{Peppi2018} Peppi, M., A. Marie, C. Belline and J.T. Borenstein. 2018.  Intracochlear drug delivery systems: a novel approach whose time has come.  {\it Expert Opin. Drug Deliv}. 15: 319--324.

\bibitem{Tandon2016} Tandon, V. , W.S. Kang, T.A. Robbins, A.J. Spencer, E.S. Kim, M.J. McKenna, S.G. Kujawa, J. Fiering, E.E. Pararas, M.J. Mescher and W.F. Sewell. 2016.  Microfabricated reciprocating micropump for intracochlear drug delivery with integrated drug\/fluid storage and electronically controlled dosing. {\it Lab Chip}.16(5): 829--846. \href{https://www.ncbi.nlm.nih.gov/pmc/articles/PMC4766044/}{https://www.ncbi.nlm.nih.gov/pmc/articles/PMC4766044/}.  Accessed on 20 August 2018.


\bibitem{WHO2018} World Health Organization, Deafness and hearing loss. 2018. \href{http://www.who.int/news-room/fact-sheets/detail/deafness-and-hearing-loss}{http://www.who.int/news-room/fact-sheets/detail/deafness-and-hearing-loss}. 15 March 2018. Accessed on 20 August 2018.


%\bibitem{Sewell2009} W.F. Sewell, J.T. Borenstein, Z. Chen, J. Fiering, O. Handzel, M. Holmboe, E.S. Kim, S.G. Kujawa, M.J. McKenna, M.M. Mescher, B. Murphy, E.E.L. Swan, M. Peppi and S. Tao, Development of a microfluidics-based intracochlear drug delivery device, {\it Audiol. Neurootol.}, 14, 411--422, 2009.

\end{thebibliography}


\newpage
%
% You only need to change the NAME OF SCENARIO in the next line and delete the footnote.
% See above for instructions on toggling from Teacher Version to Student Version.
%
\title{\ifnum\TeacherEdition=1 TEACHER VERSION \else STUDENT VERSION \fi\\Inner Ear Drug Delivery for Treating Hearing Loss}



\markboth{Intracochlear Drug Delivery}{Intracochlear Drug Delivery}

\author{Jue Wang\\
Union College\\
Schenectady NY 12308 USA\\
wangj@union.edu
}

\makeStitlePDFLaTex


\begin{abstract}
Hearing loss is difficult to treat due to the inner ear location and structure. Drawing from this challenging case, this scenario guides students to transform a treatment protocol into a mathematical model. Students engage in pre-clinical studies to examine local drug delivery to the cochlea. The delivery system is modeled as a liquid mixing problem. Students formulate the differential equation, and solve the equation using separation of variables or integrating factor. They compute the amount and concentration of drug in the cochlea over time, and help design safe and effective devices. Through observation and interpretation of solutions in a real-life scenario, students enhance their understanding and application of a differential equation model.
\end{abstract}


\section*{SCENARIO DESCRIPTION}
Over 5\% of the world's population -- or 466 million people -- has disabling hearing losses \cite{WHO2018}. The World Health Organization estimates that by 2050 over 900 million people -- or one in every ten people -- will have a disabling hearing loss.

The inner ear is surrounded by dense temporal bone and protected by the blood-cochlea barrier, as illustrated in Figure 1. The cochlea,  with the shape of a snail,  is the part of the inner ear involved in hearing. It is lined by sensory hair cells and is filled with fluid (about 0.2 milliliters, or 200 microliters). The cochlea is a particularly challenging target for drug therapy in treating hearing loss. Oral medications and injections are typically blocked by the blood-cochlea barrier, and thus ineffective in reaching or precisely dosing drugs to the cochlea.

\begin{figure}[htb]
\includegraphics[width=3.5in]{ear_anatomy2}
\centering
\caption{Human ear anatomy (adapted from \cite{Ear}).}
\end{figure}

As an alternative to systemic administration, local drug delivery methods have emerged. Intracochlear delivery releases drugs directly to the inner ear in order to establish regeneration of the sensory hair cells and auditory nerve inside the cochlea. This enables precise targeting of drug concentrations within the therapeutic window for extended delivery. Reciprocating perfusion systems based on microfluidic technologies are showing promise for direct intracochlear delivery in order to restore hearing \cite{Tandon2016}. A prototype of such system is shown in Figure 2. The implantable device is connected via a small tube to the cochlea. A battery-powered micropump pulses precise quantities of drug from a small reservoir into the cochlea in a ``push-pull'' mode, i.e. infusing and withdrawing cochlea fluid in a cyclic manner nearly simultaneously so that the fluid volume inside the cochlea stays constant.

\begin{figure}[htb]
\includegraphics[width=2.5in]{ICDD}
\centering
\caption{Delivery chip component of an intracochlear drug delivery device comprising microfluidic drug storage and flow control \cite{Peppi2018}.}
\end{figure}

In order to avoid damage to hearing structures, limitations on the maximum rate at which fluid can be pumped into the cochlea place stringent requirements on the system. It is challenging to design reliable systems that are capable of maintaining control over drug concentrations for long-term drug release. To address the difficulties in drug delivery and achieve safety and efficiency, {\bf we need your help to establish effective models for pre-clinical studies}.

\begin{enumerate}
\item {\bf Think} \\
Based on previous background information, let's see if you can come up with an appropriate differential equation model for this system. \\ (Hint 1: ``The micropump is infusing and withdrawing inner ear fluids in a cyclic manner nearly simultaneously so that the fluid volume inside the cochlea stays constant.'') \\ (Hint 2: Think about liquid mixing in the cochlea with the same flow rate in and out.) \\

\item {\bf Scenario} \\ A patient is implanted with a reciprocating perfusion device to treat hearing loss. Suppose that the drug reservoir is primed with a drug solution at a concentration of 1.2 $\mu g/\mu L$ (microgram per microliter). 1 $\mu L$ of the drug solution is infused to the patient's cochlea every 30 minutes. Simultaneously the solution is withdrawn from the cochlea at the same rate. The fluid volume inside the cochlea stays at 200 $\mu L$. Assume that the drug is delivered at a steady rate for simplicity. How much drug will be in the cochlea after a week? After two weeks? What is the concentration of the drug in the cochlea?

\begin{enumerate}
\item Sketch a schematic liquid-mixing diagram for this model. Label the given numbers on your diagram.
\item Before setting up the differential equation, define your variables clearly including units.
\item What is the rate in? What is the rate out? Include the units.
\item Set up an initial value problem for your model.
\item Solve the differential equation and find out the amount and concentration of drug in the cochlea over time. 
\item Find out the amount and concentration of drug in the cochlea after a week, and after two weeks. What do you observe? Explain. 
\item Plot the amount and concentration of drug in the cochlea over time. What do you observe? Explain.
\end{enumerate}
\smallskip

\item {\bf Design} \\ You are in charge of designing a device for long-term drug release at a reduced delivery rate, so that after 60 days the cochlea would reach 90\% of the concentration of drug solution,  i.e. 1.08 $\mu g/\mu L$. The other conditions remain the same as before. What should the delivery rate be set at?

\end{enumerate}

%A schematic representation of the drug delivery device is illustrated in Figure 3. The drug from the reservoir is first mixed with cochlea fluid in the fluid storage capacitor (mixing chamber) to maintain a constant concentration infusing into the cochlear.
%\begin{figure}[htb]
%\includegraphics[width=3.0in]{pump}
%\centering
%\caption{The drug delivery is achieved by a `push-pull' mode of fluid flow \cite{Sewell2009}.}
%\end{figure}


%\section*{Submission Version}

%For submission to the manuscript management system, \url{https://simiode.expressacademic.org} authors should submit a Teacher Version with all author identifying information stripped out. This should be a pdf version of the material.

%Authors may also submit supporting documents such as pdf's of code files, e.g., MatLab, Mathematica, Maple, SAGE, etc. and spreadsheets and data files.


%Below is a set of References which appears in the Student Version. Note, there is a second set of References one can use for the Teacher Version.



\teacher{
\section*{NOTES FOR TEACHERS}
This scenario is designed for an introductory differential equations course and can be  used as an individual assignment early in the course. Students may spend 2-3 hours on this project. After a typical problem involving salt or sugar mixing in a tank is introduced, this real-life challenge may be assigned to engage students and help them transfer learning. Discussions may be carried out in class for ``Part 1 - Think'' to stimulate ideas and lead to model formulation. 

\subsection*{Partial Solutions:}

Let $y(t)$ be the amount of drug in the cochlea of the ear at time $t$.

\begin{enumerate}
\item[] 2(d) $$\frac{dy}{dt}=\frac{1.2}{30}-\frac{y}{6000}, \;\; y(0)=0 \quad \textrm{if using } \; \mu g/min$$
$$\frac{dy}{dt}=\frac{1.2}{0.5}-\frac{y}{100}, \;\; y(0)=0 \quad \textrm{if using } \; \mu g/hr$$

\item[] 2(e) $$y(t)=240-240e^{-t/6000} \quad \textrm{if using } \; min$$
$$y(t)=240-240e^{-t/100} \quad \textrm{if using } \; hr$$

\item[] 2(f) Concentration = 0.98 $\mu g/\mu L$ after a week, 1.16 $\mu g/\mu L$ after two weeks.

\item[] 3. $$\frac{dy}{dt}=1.2 r-\frac{y}{200} r, \;\; y(0)=0$$
$$y(t)=240-240e^{-rt/200}$$
\centering{Solve for the delivery rate $r$ according to the design.}

\end{enumerate}

%\begin{thebibliography}{99}
%\end{thebibliography}
}


\end{document}
